Author's response to reviews

Title: Follow up of patients who start treatment with antidepressants: treatment satisfaction, treatment compliance, efficacy and safety

Authors:

Jesus Lopez-Torres Hidalgo (jesusl@sescam.org)
Ignacio Párraga Martínez (iparraga@sescam.jccm.es)
José María del Campo del Campo (jmariadc@sescam.jccm.es)
Alejandro Villena Ferrer (alejandrov@sescam.jccm.es)

Version: 3 Date: 11 August 2012

Author's response to reviews:

Albacete, Aug 11th, 2012

Editor of BMC Psychiatry

Dear Colleague:

I send you a new version of the manuscript entitled "Follow up of Treatment with Patients who start antidepressants: treatment satisfaction, treatment compliance, efficacy and safety" which includes the following:

- Statement in "Acknowledgements" of the financial help received from the Foundation for Healthcare Research in Castilla-La Mancha (FISCAM) (Resolution dated December 30th, 2008, reference number PI-2008/43 Official Journal of Castilla-La Mancha No. 4, January 8th, 2009). In attachments I send you the documentation that proves this governmental funding. This aid was 8,568 euros.

- Statement in "Competing interests" of financial help for the study received from Lundbeck SA Spain (commercial organization). This aid was 27,000 euros.

- Ethics Approval. This project was approved by the Research Ethics Committee of the University Hospital of Albacete on 17 April 2008. In an attachment I send you the supporting documentation.

There are no manuscripts under submission with other journals

I look forward to your reply.

Sincerely,

Jesús López-Torres Hidalgo

Doctor of Medicine. Specialist in Family and Community Medicine. Health Care Service of Castilla-La Mancha (SESCAM). Associate Professor at the Faculty of
Medicine, University of Castilla-La Mancha, Spain.